Author’s response to reviews

Title: Effect of Iron Deficiency Anemia on HbA1c in Diabetic Patients at Tikur Anbessa Specialized Teaching Hospital, Addis Ababa Ethiopia

Authors:
Absra Solomon (absrasolomon66@gmail.com)
Mintewab Hussein (mintewabhussein@gmail.com)
Mikias Negash (mikiasn2@gmail.com)
Abdurezak Ahmed (abdurezakahmed@gmail.com)
Fitsum Bekele (fbt1019@yahoo.com)
Daniel Kahase (danielasalfew@gmail.com)

Version: 3 Date: 24 Aug 2018

Author’s response to reviews:

Karen Cordovil, M.E. (Reviewer 2): The Authors,

1) I think that in methods session need the the part of Study Design. Which type of study?
   Classification the design of the study

Dear Karen Cordovil, thanks for pointing an important point. we greatly appreciate your effort in nourishing our paper. we have mentioned the study design (page 4 section-Method)

2) Please, the questionnaire need to be disponible, but it isn't in annex session. Please, talk more about of questionnaire details

   Done accordingly( Method section ---questionnaire and clinical examination-page 5)

3) Review the central aim of the study and methods analysis. The authors speak of effect measurement but don't make this.

   E.g. the Pearson correlation is a measurement of correlation, doesn't from effect.

Thanks for the information. We have adjusted accordingly (Method-data analysis page 5-6).